Author’s response to reviews

Title: Personality disorders do not influence the outcomes of chronic HCV infection treatment in Spanish prisoners. Perseo Study.

Authors:

Andrés Marco (amarco@aspb.cat)
José J Antón (quinoab@gmail.com)
Joan Trujols (itruiols@santpau.cat)
Pablo Saiz de la Hoya (pabloszh@coma.es)
José de Juan (vivaperejil1@gmail.com)
Inmaculada Faraco (ifaraco60@gmail.com)
Joan A Caylà (jcayla@aspb.cat)

Version: 3 Date: 13 February 2015

Author’s response to reviews:

In this version the line numbering is included.